Author's response to reviews

Title: The use of a silicone-coated acrylic vaginal stent in Mc Indoe vaginoplasty

Authors:

Ayhan Coskun MD (acoskun@ksu.edu.tr)
Yusuf KENAN Coban MD (cobanyk@ksu.edu.tr)
Mehmet ALI Vardar MD (vardarma@cu.edu.tr)
A. CEMIL Dalay MD (dalayc@cu.edu.tr)

Version: 2 Date: 2 May 2007

Author's response to reviews:

Here, our answers to the reviewers:

Rev.1: Although our serie is small, this is the first application of a silicone coated stent in Mc indoe vaginoplasty. This is a preliminary report.
Rev.2: Inner part of the stent is solid. This makes an effective stenting. Outer surface is coated with soft silicone. This prevents undue pressure to grafted area of newly created vagina. Main advantage of this stent is that it is cheaper than other materials. It is very comfortable for patients.
Rev.3: Unfortunately, we did not reach an information about presence of our newly designed stent in past.
Rev.4: Thanks for your considerations. We will take your thoughts into account in our future studies. The patients have declared to have comfort with intercourse. This make us to write this manuscript.